Author's response to reviews

**Title:** Quality assessment of DNA derived from up to 30 years old formalin fixed paraffin embedded tissue (FFPE) for PCR-based methylation analysis using SMART-MSP and MS-HRM

**Authors:**

Lasse S Kristensen (lasse@humgen.au.dk)
Tomasz K Wojdacz (wojdacz@humgen.au.dk)
Britta B Thestrup (boserup@humgen.au.dk)
Carsten Wiuf (wiuf@birc.au.dk)
Henrik Hager (henrikhager@mac.com)
Lise Lotte Hansen (Lotte@humgen.au.dk)

**Version:** 5  **Date:** 14 December 2009

**Author's response to reviews:**

The manuscript has been revised according to the list provided by BMC Cancer.
1. The sentence that all authors have red and approved the manuscript is added.
2. Key words have been removed
3. Reference 4 has been updated
4. Vertical lines have been removed from the tables, except the ones separating tile headlines from the results in the tables.
5. All figures have been cropped and the guidelines has been followed.
6. The entire text has been thoroughly red, and minor corrections has been made throughout the manuscript.